\begin{table}%tA.1
\caption{\label{reactions}Reactions used in the model. The rate coefficients are given for 7~K. Reaction rates less than 10$^{-15}$~\cc\ s$^{-1}$ are not taken into account in our models. Reference~(1) corresponds to \citet{Gerlich90}, references~(2)$-$(4) correspond to Hugo, OSU~07, and this paper respectively. 
For OSU~07, branching ratios involving spin states have been infered from quantum mechanical rules. For reactions involving grains, a grain radius of 0.1~$\mu$m and a sticking coefficient of~1 have been considered. (5)~\citet{Datz95b} (6)~\citet{Datz95a}, (7)~\citet{Zhaunerchyk08}, (8)~\citet{Larsson97}, (9)~\citet{Molek07}.}
\par
%\centering
\small
\begin{tabular}{lclclclccc}
\hline
\hline
&&&Reaction                         &         &  &         &   Rate           &  $\alpha$               & Reference \\
&&	           &              &         &  &         &                  &   cm$^3$~s$^{-1}$  & \\
\hline
%&&& \\[-8pt]
H$^+$ &+& o-\hdb     &$\rightarrow$ & H$^+$   & +&  p-\hdb &   & 2.2 $\times$ 10$^{-10}$ & (1)  \\
p-\htpb  &+&  o-\hdb &$\rightarrow$ & p-\htpb & +&  p-\hdb  &  & 3.3 $\times$ 10$^{-10}$ & (2)  \\ 
\qquad&&\qquad       &$\rightarrow$ & o-\htpb & + & p-\hdb & $k0_{\rm poop}$ & 3.9 $\times$ 10$^{-10}$ & (2)\\
 \qquad&&\qquad      &$\rightarrow$ & o-\htpb & +&  o-\hdb & $k0_{\rm pooo}$ & 7.8 $\times$ 10$^{-12}$ & (2)\\
o-\htpb & + & o-\hdb &$\rightarrow$ & p-\htpb & + & p-\hdb & $k0_{\rm oopp}$ & 1.0 $\times$ 10$^{-10}$ & (2)\\
 \qquad&&\qquad      &$\rightarrow$ & p-\htpb & + & o-\hdb &  $k0_{\rm oopo}$ & 4.1 $\times$ 10$^{-10}$ &  (2)\\
 \qquad&&\qquad      &$\rightarrow$ & o-\htpb & + & p-\hdb &  $k0_{\rm ooop}$ & 1.0 $\times$ 10$^{-10}$ &  (2)\\
p-\htpb & + & HD     &$\rightarrow$ & o-\htpb & + & HD  &  $k1_{\rm pdod}$ & 8.2 $\times$ 10$^{-13}$ &  (2)\\
 \qquad&&\qquad      &$\rightarrow$ & p-\hddp & + & p-\hdb   &  $k1_{\rm pdpp}$ & 3.2 $\times$ 10$^{-10}$ &  (2)\\  
 \qquad&&\qquad      &$\rightarrow$ & p-\hddp & + & o-\hdb   &  $k1_{\rm pdpo}$ & 4.3 $\times$ 10$^{-10}$ & (2)\\  
 \qquad&&\qquad      &$\rightarrow$ & o-\hddp & + & p-\hdb   &  $k1_{\rm pdop}$  & 6.8 $\times$ 10$^{-10}$ &  (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ & o-\hddp & + & o-\hdb   &  $k1_{\rm pdoo}$ & 1.6 $\times$ 10$^{-11}$ &  (2)\\ 
o-\htpb & + & HD     &$\rightarrow$ & p-\hddp & + & o-\hdb   &  $k1_{\rm odpo}$ & 1.7 $\times$ 10$^{-10}$ &  (2)\\
 \qquad&&\qquad      &$\rightarrow$ & p-\htpb & + & HD  &  $k1_{\rm odpd}$ & 4.35 $\times$ 10$^{-11}$ &  (2)\\
 \qquad&&\qquad      &$\rightarrow$ & o-\hddp & + & p-\hdb   &  $k1_{\rm odop}$ & 2.3 $\times$ 10$^{-10}$ &  (2)\\
 \qquad&&\qquad      &$\rightarrow$ & o-\hddp & + & o-\hdb   &  $k1_{\rm odoo}$ & 1.0 $\times$ 10$^{-9}$ & (2)\\
p-\hddpb & + &o-\hdb &$\rightarrow$ & o-\hddpb& + & p-\hdb   &  & 1.3 $\times$ 10$^{-9}$ & (2)\\ 
\qquad&&\qquad	   &$\rightarrow$ & p-\htpb & + & HD   &   $k\_1_{\rm popd}$ & 3.9 $\times$ 10$^{-14}$  & (2)\\
\qquad&&\qquad       &$\rightarrow$ & o-\hddpb& + & o-\hdb   &   & 2.0 $\times$ 10$^{-15}$  & (2)\\ 
o-\hddpb & +& o-\hdb &$\rightarrow$ & p-\htpb & + & HD   & $k\_1_{\rm oopd}$ & 7.7 $\times$ 10$^{-11}$ & (2)\\
 \qquad&&\qquad      &$\rightarrow$ & o-\htpb & + &HD   &  $k\_1_{\rm oood}$ & 8.3 $\times$ 10$^{-11}$  & (2)\\
 \qquad&&\qquad      &$\rightarrow$ & p-\hddpb& + & p-\hdb   & &  9.2 $\times$ 10$^{-11}$  & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ & p-\hddpb& + &o-\hdb   & & 1.8 $\times$ 10$^{-10}$  & (2)\\
 \qquad&&\qquad      &$\rightarrow$ & o-\hddpb& + &p-\hdb   & &  2.35 $\times$ 10$^{-10}$  & (2)\\ 
o-\hddpb & +& p-\hdb &$\rightarrow$ & p-\hddpb& + &o-\hdb   & &  4.2 $\times$ 10$^{-15}$  & (2)\\ 
o-\hddpb & + & HD    &$\rightarrow$ & p-\htpb & + &  o-\ddb  & &  1.2 $\times$ 10$^{-15}$    & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ & p-\hddpb& + & HD   & &   5.4 $\times$ 10$^{-11}$  & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ & p-\ddhpb& + &  p-\hdb  & &   7.3 $\times$ 10$^{-11}$    & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ & p-\ddhpb& + &  o-\hdb  & &  3.8 $\times$ 10$^{-10}$   & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ & o-\ddhpb& + &  p-\hdb  & &   1.8 $\times$ 10$^{-10}$    & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ & o-\ddhpb& + &  o-\hdb  & &   6.8 $\times$ 10$^{-10}$    & (2)\\ 
 p-\hddpb & + & HD   &$\rightarrow$ & p-\ddhpb& + &  p-\hdb  & &   4.8 $\times$ 10$^{-10}$   & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ & p-\ddhpb& + &  o-\hdb  & &   2.1 $\times$ 10$^{-12}$    & (2)\\
 \qquad&&\qquad      &$\rightarrow$ & o-\ddhpb& + &  p-\hdb  & &   7.5 $\times$ 10$^{-10}$    & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ & o-\ddhpb& + & o-\hdb  & &   2.3 $\times$ 10$^{-10}$    & (2)\\ 
 p-\ddhpb& +& o-\hdb &$\rightarrow$ & p-\hddpb& + &  HD   & &   3.1 $\times$ 10$^{-11}$  & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ & p-\ddhpb& + &  p-\hdb   & &   3.3 $\times$ 10$^{-10}$  & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ & o-\hddpb& + & HD   & &   1.2 $\times$ 10$^{-13}$  & (2)\\ 
o-\ddhpb & +& o-\hdb &$\rightarrow$ & p-\hddpb& + & HD   & &   3.8 $\times$ 10$^{-12}$  & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ & o-\ddhpb& + & p-\hdb   & &   4.45 $\times$ 10$^{-10}$  & (2)\\ 
p-\ddhpb & + & HD    &$\rightarrow$ & p-\hddpb& + &   o-\ddb  & &   2.8 $\times$ 10$^{-15}$    & (2)\\
  \qquad&&\qquad     &$\rightarrow$ & o-\ddhpb& + &  HD  & &   9.1 $\times$ 10$^{-11}$    & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ &  p-\dtpb& + &  p-\hdb  & &   3.3 $\times$ 10$^{-11}$    & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ & p-\dtpb & + & o-\hdb  & &   9.75 $\times$ 10$^{-11}$    & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ &  m-\dtpb& + &  p-\hdb  & &  2.6 $\times$ 10$^{-10}$    & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ &  m-\dtpb& + &  o-\hdb  & &   7.8 $\times$ 10$^{-10}$    & (2)\\ 
 o-\ddhpb & + &  HD  &$\rightarrow$ & m-\dtpb & + & p-\hdb  & &  1.6 $\times$ 10$^{-10}$    & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ & m-\dtpb & + &  o-\hdb  & &   4.5 $\times$ 10$^{-10}$    & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ & o-\dtpb & + &   p-\hdb  & &   1.9 $\times$ 10$^{-10}$    & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ & o-\dtpb & + & o-\hdb  & &   4.7 $\times$ 10$^{-10}$    & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ & p-\ddhpb& + &  HD  & &   9.9 $\times$ 10$^{-14}$    & (2)\\ 
m-\dtpb&  +& o-\hdb  &$\rightarrow$ & o-\ddhpb& + & HD   & &   1.3 $\times$ 10$^{-11}$  & (2)\\ 
 \qquad&&\qquad      &$\rightarrow$ & p-\ddhpb& + & HD   & &   1.9 $\times$ 10$^{-14}$  & (2)\\ 
o-\dtpb & +& o-\hdb  &$\rightarrow$ & o-\ddhpb& + & HD   & &   4.4 $\times$ 10$^{-14}$  & (2)\\ 
p-\dtpb &+ & o-\hdb  &$\rightarrow$ & p-\ddhpb& + &  HD   & &   2.6 $\times$ 10$^{-13}$  & (2)\\ 
p-\dtpb &  +&   HD   &$\rightarrow$ & m-\dtpb & + &  HD   & &  7.75 $\times$ 10$^{-10}$  & (2)\\ 
m-\dtpb&  + & HD     &$\rightarrow$ & o-\dtpb & + &  HD   & &   2.5 $\times$ 10$^{-10}$  & (2)\\ 
%&&& \\[-8pt] 
\hline
 \qquad&&\qquad          &$\rightarrow$&  p-\dtpb  &+ &  HD   & &  9.5 $\times$ 10$^{-12}$  & (2)\\ 
o-\dtpb  & + &  HD       &$\rightarrow$&  m-\dtpb   &+ &  HD   & &   8.45 $\times$ 10$^{-13}$  & (2)\\
o-\htpb  & + &  o-\ddb   &$\rightarrow$&   o-\hddpb &  + &  HD & &  2.1 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&  o-\ddhpb   &+ &  o-\hdb & &   1.3 $\times$ 10$^{-9}$ & (2)\\ 
o-\htpb  & + &  p-\ddb   &$\rightarrow$&   o-\hddpb  & + &  HD & &   5.0 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   p-\ddhpb  & + &  o-\hdb & &   9.7 $\times$ 10$^{-10}$ & (2)\\ 
p-\htpb  & + &  o-\ddb   &$\rightarrow$&   p-\hddpb  & + &  HD & &  1.6 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   o-\hddpb  & + &  HD & &  1.4 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&  o-\ddhpb  & + &  p-\hdb & &   5.95 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   o-\ddhpb &  + &  o-\hdb & &   6.2 $\times$ 10$^{-10}$ & (2)\\ 
p-\htpb  & + &  p-\ddb   &$\rightarrow$&   p-\hddpb  & + &  HD & &   2.55 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   o-\hddpb  & + &  HD & &   3.45 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&  p-\ddhpb  & + &  p-\hdb & &   4.0 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   p-\ddhpb  & +&   o-\hdb & &   4.9 $\times$ 10$^{-10}$ & (2)\\ 
o-\hddpb &  + &  o-\ddb  &$\rightarrow$&   p-\ddhpb  & + &  HD & &   1.7 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&  o-\ddhpb  & + &  HD & &  3.9 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&  m-\dtpb & +  & o-\hdb & &   3.2 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   o-\dtpb &  + &  o-\hdb & &   5.3 $\times$ 10$^{-10}$ & (2)\\ 
o-\hddpb &  + &  p-\ddb  &$\rightarrow$&   o-\hddpb &  + &  o-\ddb & &   4.8 $\times$ 10$^{-11}$ & (2)\\ 
\qquad&&\qquad           &$\rightarrow$&  p-\ddhpb  & + &  HD & &  4.1 $\times$ 10$^{-10}$ & (2)\\  
 \qquad&&\qquad          &$\rightarrow$&  o-\ddhpb  & + &  HD & &   3.6 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&  p-\dtpb  & + &  o-\hdb & &   6.1 $\times$ 10$^{-11}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&  m-\dtpb  & + &  o-\hdb & &   5.05 $\times$ 10$^{-10}$ & (2)\\ 
p-\hddpb &  + &  o-\ddb  &$\rightarrow$&   p-\ddhpb &  + &  HD & &   1.2 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   o-\ddhpb  & + &  HD & &   2.7 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&  m-\dtpb  & + &  p-\hdb & &   4.3 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&  o-\dtpb  & + &  p-\hdb & &  5.8 $\times$ 10$^{-10}$ & (2)\\ 
p-\hddpb &  + &  p-\ddb  &$\rightarrow$&   p-\hddpb &  +&   o-\ddb & &  6.55 $\times$ 10$^{-11}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&  p-\ddhpb  & +  & HD & &  4.1 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   o-\ddhpb &  + &  HD & &  2.8 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   p-\dtpb  & + &  p-\hdb & &  6.9 $\times$ 10$^{-11}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   m-\dtpb &  +  & p-\hdb & &  5.8 $\times$ 10$^{-10}$ & (2)\\ 
o-\ddhpb &  +   &o-\ddb  &$\rightarrow$&   p-\ddhpb  & +&  o-\ddb & &  3.85 $\times$ 10$^{-14}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   m-\dtpb  & + &  HD & &  5.7 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   o-\dtpb  & + &  HD & &  6.4 $\times$ 10$^{-10}$ & (2)\\ 
o-\ddhpb  & + &  p-\ddb  &$\rightarrow$&   p-\ddhpb  & +&   p-\ddb & &  2.1 $\times$ 10$^{-14}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&  p-\ddhpb   &+ &  o-\ddb & &  1.6 $\times$ 10$^{-10}$ & (2)\\  
 \qquad&&\qquad          &$\rightarrow$&   o-\ddhpb  & +&   o-\ddb & &  8.8 $\times$ 10$^{-11}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   p-\dtpb   &+ &  HD & &  8.4 $\times$ 10$^{-11}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   m-\dtpb  & + &  HD & &  6.4 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   o-\dtpb  & + &  HD & &  3.5 $\times$ 10$^{-10}$ & (2)\\ 
p-\ddhpb  & + &  o-\ddb  &$\rightarrow$&   o-\ddhpb &  +&   p-\ddb & &  5.0 $\times$ 10$^{-13}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&  o-\ddhpb  & + &  o-\ddb & &  3.7 $\times$ 10$^{-11}$ & (2)\\  
 \qquad&&\qquad          &$\rightarrow$&   p-\dtpb  & + &  HD & &  7.7 $\times$ 10$^{-11}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&  m-\dtpb   &+ &  HD & &  8.6 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   o-\dtpb  & + &  HD & &  3.05 $\times$ 10$^{-10}$ & (2)\\ 
p-\ddhpb &  + &  p-\ddb  &$\rightarrow$&  p-\ddhpb & + & o-\ddb & & 3.7 $\times$ 10$^{-11}$ & (2)\\
 \qquad&&\qquad          &$\rightarrow$&  o-\ddhpb & + & p-\ddb & & 2.0 $\times$ 10$^{-11}$ & (2)\\
 \qquad&&\qquad          &$\rightarrow$&  o-\ddhpb  & + &  o-\ddb & &  8.5 $\times$ 10$^{-11}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   p-\dtpb &  + &  HD & & 1.15 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   m-\dtpb &  + &  HD & &  9.5 $\times$ 10$^{-10}$ & (2)\\ 
m-\dtpb &  +  & o-\ddb   &$\rightarrow$&   p-\dtpb &  + &  o-\ddb & &  8.6 $\times$ 10$^{-12}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   m-\dtpb &  + &  p-\ddb & &  1.7 $\times$ 10$^{-15}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&  o-\dtpb  & + &  p-\ddb & &  1.2 $\times$ 10$^{-12}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   o-\dtpb &  + &  o-\ddb & &  3.0 $\times$ 10$^{-10}$ & (2)\\ 
m-\dtpb & + & p-\ddb     &$\rightarrow$&   p-\dtpb & + &  p-\ddb & &  5.9 $\times$ 10$^{-12}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   p-\dtpb &  + &  o-\ddb & &  4.7 $\times$ 10$^{-11}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   m-\dtpb & + &  o-\ddb & &  6.1 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&  o-\dtpb  & +  & p-\ddb & &  4.7 $\times$ 10$^{-11}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   o-\dtpb &  + &  o-\ddb & &  5.1 $\times$ 10$^{-11}$ & (2)\\ 
o-\dtpb &  + &  o-\ddb   &$\rightarrow$&   m-\dtpb & + &  o-\ddb & &  1.0 $\times$ 10$^{-12}$ & (2)\\ 
 \qquad&&\qquad          &$\rightarrow$&   o-\dtpb  & +&   p-\ddb & &  1.15 $\times$ 10$^{-15}$ & (2)\\ 
o-\dtpb &  + &  p-\ddb   &$\rightarrow$&   p-\dtpb  &+ &  o-\ddb & &  9.2 $\times$ 10$^{-11}$ & (2)\\ 
%&&& \\[-8pt] 
 \hline
 \qquad&&\qquad 	        &$\rightarrow$&   m-\dtpb  & + &  p-\ddb & &  1.6 $\times$ 10$^{-13}$ & (2)\\
 \qquad&&\qquad           &$\rightarrow$&  m-\dtpb & + & o-\ddb & &  5.7 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad           &$\rightarrow$&  o-\dtpb  & +  & o-\ddb & &  1.6 $\times$ 10$^{-10}$ & (2)\\ 
 p-\dtpb &  +  & o-\ddb   &$\rightarrow$&   m-\dtpb  & +  & p-\ddb & &  2.8 $\times$ 10$^{-14}$ & (2)\\ 
 \qquad&&\qquad           &$\rightarrow$&   m-\dtpb & + & o-\ddb & &  7.0 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad           &$\rightarrow$&   o-\dtpb &  + &  p-\ddb & & 1.7 $\times$ 10$^{-11}$ & (2)\\ 
p-\dtpb  & +   &p-\ddb    &$\rightarrow$&   m-\dtpb &  + &  p-\ddb & &  4.8 $\times$ 10$^{-10}$ & (2)\\ 
 \qquad&&\qquad           &$\rightarrow$&   m-\dtpb &  + &  o-\ddb & &  4.4 $\times$ 10$^{-10}$ & (2)\\ 
o-H$_3^+$  & +  & CO      &$\rightarrow$&   HCO$^+$  &  + &  o-\hdb      & $k_{\rm CO}$   & 1.6 $\times$ 10$^{-9}$   & (3)\\
p-H$_3^+$ &  +  & CO      &$\rightarrow$&  HCO$^+$  &  + &  p-\hdb      & 1/2 $k_{\rm CO}$   & 8.1 $\times$ 10$^{-10}$   & (3)\\
\qquad&&\qquad            &$\rightarrow$&   HCO$^+$ &  + &  o-\hdb  &  1/2 $k_{\rm CO}$   & 8.1 $\times$ 10$^{-10}$   & (3)\\
o-\hddpb &  + &  CO       &$\rightarrow$&   HCO$^+$ &  + &  HD    &  2/3 $k'_{\rm CO}$$^1$   & 9.4 $\times$ 10$^{-10}$   & (3)\\
\qquad&&\qquad            &$\rightarrow$&  DCO$^+$  & + &  o-\hdb   &  1/3 $k'_{\rm CO}$   & 4.7 $\times$ 10$^{-10}$   & (3)\\
p-\hddpb  & + &  CO       &$\rightarrow$&   HCO$^+$  & + &  HD      &  2/3 $k'_{\rm CO}$   & 9.4 $\times$ 10$^{-10}$   & (3)\\
\qquad&&\qquad            &$\rightarrow$&   DCO$^+$  & + &  p-\hdb    &  1/3 $k'_{\rm CO}$   & 4.7 $\times$ 10$^{-10}$   & (3)\\
o-\ddhpb  & +  & CO       &$\rightarrow$&   HCO$^+$  & + & o-D$_2$    &  1/3 $k''_{\rm CO}$   & 4.3 $\times$ 10$^{-10}$   & (3)\\
\qquad&&\qquad            &$\rightarrow$&  DCO$^+$  & +  & HD    &  2/3 $k''_{\rm CO}$   & 8.6 $\times$ 10$^{-10}$   & (3)\\
p-\ddhpb  & + &  CO       &$\rightarrow$&  HCO$^+$  & +  & p-D$_2$      &  $1/3 k''_{\rm CO}$   & 4.3 $\times$ 10$^{-10}$   & (3)\\
\qquad&&\qquad            &$\rightarrow$&   DCO$^+$ &  + &  HD   &  2/3 $k''_{\rm CO}$   & 8.6 $\times$ 10$^{-10}$   & (3)\\
o-\dtpb &  +  & CO        &$\rightarrow$&   DCO$^+$  & + &  o-D$_2$      &  $k'''_{\rm CO}$   & 1.2 $\times$ 10$^{-9}$   & (3)\\
m-\dtpb &  +  & CO        &$\rightarrow$&   DCO$^+$  & + &  o-D$_2$      &  1/2 $k'''_{\rm CO}$   & 6.0 $\times$ 10$^{-10}$   & (3)\\
\qquad&&\qquad            &$\rightarrow$&   DCO$^+$  & + &  p-D$_2$     &  1/2 $k'''_{\rm CO}$   & 6.0 $\times$ 10$^{-10}$   & (3)\\
p-\dtpb  & +  & CO        &$\rightarrow$&   DCO$^+$  & +  & p-D$_2$      &  $k'''_{\rm CO}$   & 1.2 $\times$ 10$^{-9}$   & (3)\\
p-\htpb &  +  & N$_2$     &$\rightarrow$&   N$_2$H$^+$  & +&   o-H$_2$  &    1/2 $k_{N_2}$ & 8.5 $\times$ 10$^{-10}$ & (3)\\
\qquad&&\qquad            &$\rightarrow$&    N$_2$H$^+$ &  +&  p-H$_2$  &     1/2 $k_{N_2}$ & 8.5 $\times$ 10$^{-10}$ & (3)\\
o-\htpb  & +  & N$_2$     &$\rightarrow$&   N$_2$H$^+$  & + &  o-H$_2$      & $k_{N_2}$ & 1.7 $\times$ 10$^{-9}$ & (3)\\
o-\hddpb  & +  & N$_2$    &$\rightarrow$&   N$_2$H$^+$  & + & HD  &   2/3 $k'_{N_2}$$^1$& 1.0 $\times$ 10$^{-9}$ & (3)\\
\qquad&&\qquad            &$\rightarrow$&  N$_2$D$^+$  & + &  o-H$_2$ &  1/3 $k'_{N_2}$ & 5.0 $\times$ 10$^{-10}$ & (3)\\
p-\hddpb  & + &  N$_2$    &$\rightarrow$&   N$_2$H$^+$  & + &  HD  &   2/3 $k'_{N_2}$ & 1.0 $\times$ 10$^{-9}$ & (3)\\
\qquad&&\qquad            &$\rightarrow$&   N$_2$D$^+$  & + &  p-H$_2$  &   1/3 $k'_{N_2}$ & 5.0 $\times$ 10$^{-10}$ & (3)\\
o-\ddhpb &  + &  N$_2$    &$\rightarrow$&   N$_2$H$^+$  & + &  p-D$_2$  &  1/3 $k''_{N_2}$ & 4.5 $\times$ 10$^{-10}$ & (3)\\
\qquad&&\qquad            &$\rightarrow$&   N$_2$D$^+$  & + &  HD  &   2/3 $k''_{N_2}$ & 9.1 $\times$ 10$^{-10}$ & (3)\\
p-\ddhpb  & +  & N$_2$    &$\rightarrow$&   N$_2$H$^+$  & + &  p-D$_2$  &   1/3 $k''_{N_2}$ & 4.5 $\times$ 10$^{-10}$ & (3)\\
\qquad&&\qquad            &$\rightarrow$&   N$_2$D$^+$  & + &  HD  &    2/3 $k''_{N_2}$ & 9.1 $\times$ 10$^{-10}$ & (3)\\
o-\dtpb  & + &  N$_2$     &$\rightarrow$&   N$_2$D$^+$  & + &  o-D$_2$  &  $k'''_{N_2}$ & 1.3 $\times$ 10$^{-9}$ & (3)\\
m-\dtpb  & + &  N$_2$     &$\rightarrow$&   N$_2$D$^+$  & + &  o-D$_2$  &  1/2 $k'''_{N_2}$ & 6.3 $\times$ 10$^{-10}$ & (3)\\
\qquad&&\qquad            &$\rightarrow$&   N$_2$D$^+$  & + &  p-D$_2$  &  1/2 $k'''_{N_2}$ & 6.3 $\times$ 10$^{-10}$ & (3)\\
p-\dtpb  & + &  N$_2$     &$\rightarrow$&   N$_2$D$^+$  & +&  p-D$_2$  &  $k'''_{N_2}$ & 1.3 $\times$ 10$^{-9}$ & (3)\\
H$^+$ & +  & e$^-$       &$\rightarrow$&     H            &        &    &  $k_{\rm rec}$  &  4.9 $\times$ 10$^{-11}$& (3) \\
o-H$_3^+$  & + &  e$^-$   &$\rightarrow$&   H  &+ & H\qquad\qquad +\quad H & 3/4 o-$k_{\rm rec1}$ &   3.6 $\times$ 10$^{-8}$ & (4, 5) \\
\qquad&&\qquad            &$\rightarrow$&   o-\hdb & + & H &  1/4 o-$k_{\rm rec1}$ &   1.2 $\times$ 10$^{-8}$ &  (4, 5)\\
p-H$_3$$^+$  & +&   e$^-$ &$\rightarrow$&    H&+&H\qquad\qquad +\quad H & 3/4 p-$k_{\rm rec1}$ &   5.6 $\times$ 10$^{-7}$ & (4, 5) \\
\qquad&&\qquad            &$\rightarrow$&    p-\hdb& + &H & 1/8 p-$k_{\rm rec1}$ &   9.4 $\times$ 10$^{-8}$ & (4, 5) \\
\qquad&&\qquad            &$\rightarrow$&   o-\hdb &+ &H & 1/8 p-$k_{\rm rec1}$ &   9.4 $\times$ 10$^{-8}$ & (4, 5) \\
o-H$_2$D$^+$ & + & e$^-$  &$\rightarrow$&   H& + &H\qquad\qquad +\quad D &3/4 o-$k_{\rm rec2}$ & 2.6 $\times$ 10$^{-7}$ &(4, 6) \\
\qquad&&\qquad            &$\rightarrow$&  o-\hdb & + & D &  $\sim$0.07 o-$k_{\rm rec2}$ & 2.5 $\times$ 10$^{-8}$ & (4, 6)\\
\qquad&&\qquad            &$\rightarrow$&   HD&  + & D &  $\sim$0.18 o-$k_{\rm rec2}$ & 6.25 $\times$ 10$^{-8}$ &(4, 6) \\
p-H$_2$D$^+$  &+ &e$^-$   &$\rightarrow$&   H & + & H\qquad\qquad +\quad D & 3/4 p-$k_{\rm rec2}$ & 1.8 $\times$ 10$^{-7}$ &(4, 6) \\
\qquad&&\qquad            &$\rightarrow$&   p-\hdb & + & D &  $\sim$0.07 p-$k_{\rm rec2}$ & 1.7 $\times$ 10$^{-8}$ &(4, 6) \\
\qquad&&\qquad            &$\rightarrow$&   HD & + & D &  $\sim$0.18 p-$k_{\rm rec2}$ & 4.3 $\times$ 10$^{-8}$ &(4, 6) \\
o-D$_2$H$^+$ & + &e$^-$   &$\rightarrow$&   H & + & D\qquad\qquad +\quad D & $\sim$0.77 o-$k_{\rm rec3}$   &  4.3 $\times$ 10$^{-8}$  &(4, 7) \\
\qquad&&\qquad            &$\rightarrow$&   HD & + & D &  $\sim$0.13 o-$k_{\rm rec3}$ & 7.3 $\times$ 10$^{-9}$ & (4, 7)\\
\qquad&&\qquad            &$\rightarrow$&   o-D$_2$ &+ &H &  $\sim$0.10 o-$k_{\rm rec3}$ & 5.6 $\times$ 10$^{-9}$ &(4, 7) \\
p-D$_2$H$^+$ & +& e$^-$   &$\rightarrow$&   H & + & D\qquad\qquad +\quad D &$\sim$0.77 p-$k_{\rm rec3}$   &  4.6 $\times$ 10$^{-8}$  & (4, 7)\\
\qquad&&\qquad            &$\rightarrow$&   HD & + & D &  $\sim$0.13 p-$k_{\rm rec3}$ & 7.8 $\times$ 10$^{-9}$ &(4, 7) \\
\qquad&&\qquad            &$\rightarrow$&  p-D$_2$& +& H & $\sim$0.10 p-$k_{\rm rec3}$ & 6.0 $\times$ 10$^{-9}$ &(4, 7) \\
o-D$_3$$^+$ &  + &  e$^-$ &$\rightarrow$&   D & + & D\qquad\qquad +\quad D &3/4 o-$k_{\rm rec4}$  & 1.8 $\times$ 10$^{-7}$ & (4, 8) \\
\qquad&&\qquad            &$\rightarrow$&   o-D$_2$ & + & D &  1/4 o-$k_{\rm rec4}$ & 6.1 $\times$ 10$^{-8}$ &(4, 8) \\
p-D$_3$$^+$  &+ & e$^-$   &$\rightarrow$&    D & + & D\qquad\qquad +\quad D &3/4 p-$k_{\rm rec4}$  & 1.5 $\times$ 10$^{-7}$ & (4, 8) \\
\qquad&&\qquad            &$\rightarrow$&    p-D$_2$ &  + & D &  1/4 p-$k_{\rm rec4}$  & 5.0 $\times$ 10$^{-8}$ &(4, 8)  \\
m-D$_3$$^+$ &  + & e$^-$  &$\rightarrow$&   D & + & D\qquad\qquad +\quad D & 3/4 m-$k_{\rm rec4}$  & 3.6 $\times$ 10$^{-7}$ &(4, 8)  \\
\qquad&&\qquad            &$\rightarrow$&    o-D$_2$ &  +&  D &  1/8 m-$k_{\rm rec4}$  & 6.0 $\times$ 10$^{-8}$ & (4, 8) \\
%&&& \\[-8pt]
\hline
\qquad&&\qquad              &$\rightarrow$&    p-D$_2$   &+&  D &  1/8 m-$k_{\rm rec4}$  & 6.0 $\times$ 10$^{-8}$ & (4, 8) \\
N$_2$H$^+$  &  + &  e$^-$   &$\rightarrow$&    H   &+ &  N$_2$     & $k_{\rm N_2H^+}$ & 6.8 $\times$ 10$^{-7}$  & (9)\\
N$_2$D$^+$  &  + &  e$^-$   &$\rightarrow$&    D  & + &  N$_2$ &    $k_{\rm N_2D^+}$ &  6.8 $\times$ 10$^{-7}$ & (9)\\
HCO$^+$   & + &  e$^-$      &$\rightarrow$&    H   &+&   CO &    $k_{\rm HCO^+}$ & 3.2 $\times$ 10$^{-6}$   & (2)\\
DCO$^+$ &   + &  e$^-$      &$\rightarrow$&    D   &+ &  CO &      $k_{\rm DCO^+}$ &  3.2 $\times$ 10$^{-6}$ & (2)\\
gr$^0$ &   + & e$^-$        &$\rightarrow$&    gr$^-$ && &    $k_{\rm e}$ &   3.7 $\times$ 10$^{-3}$ & (4)\\
H$^+$ &  + &  gr$^-$        &$\rightarrow$&    H  & + &  gr$^0$&   $k_{\rm gr}$  & 6.0 $\times$ 10$^{-4}$   &  (4)\\
HD$^+$  & +&  gr$^-$       &$\rightarrow$&    H  & + &  D   \qquad\qquad +\quad    gr$^0$&  2/3 $k_{\rm gr1}$ &  2.3 $\times$ 10$^{-4}$  &  (4)\\
\qquad&&\qquad              &$\rightarrow$&    HD & + & gr$^0$&   1/3 $k_{\rm gr1}$  &  1.2 $\times$ 10$^{-4}$  &  (4)\\
o-D$_2^+$  & +  & gr$^-$    &$\rightarrow$&    D  & + &  D   \qquad\qquad +\quad   gr$^0$& 2/3 $k_{\rm gr2}$  & 2.0 $\times$ 10$^{-4}$   &  (4)\\
 \qquad&&\qquad             &$\rightarrow$&    o-D$_2$  & + &   gr$^0$&  1/3 $k_{\rm gr2}$   & 1.0 $\times$ 10$^{-4}$   &  (4)\\
p-D$_2^+$  & + &  gr$^-$    &$\rightarrow$&    D  & +  & D   \qquad\qquad +\quad    gr$^0$& 2/3 $k_{\rm gr2}$ &  2.0 $\times$ 10$^{-4}$  &  (4)\\
 \qquad&&\qquad             &$\rightarrow$&    p-D$_2$ &  + &  gr$^0$&  1/3 $k_{\rm gr2}$  & 1.0 $\times$ 10$^{-4}$   &  (4)\\
He$^+$ &  +  & gr$^-$       &$\rightarrow$&    He  & + &  gr$^0$& &  3.0 $\times$ 10$^{-4}$   &  (4)\\
p-H$_3^+$ &  + &  gr$^-$    &$\rightarrow$&    p-H$_2$  & +&   H   \qquad\qquad +\quad    gr$^0$& 1/6 $k_{\rm gr3}$ &  5.8 $\times$ 10$^{-5}$  & (4)\\
\qquad&&\qquad              &$\rightarrow$&    o-H$_2$ &  +&   H   \qquad\qquad +\quad    gr$^0$& 1/6 $k_{\rm gr3}$ & 5.8 $\times$ 10$^{-5}$   &  (4)\\
\qquad&&\qquad      	  &$\rightarrow$&    3H  & + &  gr$^0$       & 2/3 $k_{\rm gr3}$  & 2.3 $\times$ 10$^{-4}$  &  (4)\\
o-H$_3^+$ & + & gr$^-$      &$\rightarrow$&    o-H$_2$ &  + &  H   \qquad\qquad +\quad   gr$^0$& 1/3 $k_{\rm gr3}$ & 1.2 $\times$ 10$^{-4}$   & (4)\\
 \qquad&&\qquad             &$\rightarrow$&    3H  & + &  gr$^0$       & 2/3 $k_{\rm gr3}$  &  2.3 $\times$ 10$^{-4}$  &  (4)\\
p-H$_2$D$^+$ & + & gr$^-$   &$\rightarrow$&    p-H$_2$  & + &  D   \qquad\qquad +\quad   gr$^0$& 1/6 $k_{\rm gr4}$ & 5.0 $\times$ 10$^{-5}$   & (4)\\
\qquad&&\qquad              &$\rightarrow$&    HD  & + &  H  \qquad\qquad +\quad    gr$^0$& 1/3 $k_{\rm gr4}$  &  1.0 $\times$ 10$^{-4}$  &  (4)\\
 \qquad&&\qquad             &$\rightarrow$&    2H  & +  & D   \qquad\qquad +\quad   gr$^0$& 1/2 $k_{\rm gr4}$ & 1.5 $\times$ 10$^{-4}$   &  (4)\\
o-H$_2$D$^+$ &  + &gr$^-$   &$\rightarrow$&    2H  & + &  D   \qquad\qquad +\quad   gr$^0$& 1/2 $k_{\rm gr4}$ & 1.5 $\times$ 10$^{-4}$   & (4)\\
 \qquad&&\qquad             &$\rightarrow$&    o-H$_2$ &  + &  D  \qquad\qquad +\quad    gr$^0$& 1/6 $k_{\rm gr4}$ & 5.0 $\times$ 10$^{-5}$  &  (4)\\
\qquad&&\qquad              &$\rightarrow$&    HD  & + &  H   \qquad\qquad +\quad    gr$^0$& 1/3 $k_{\rm gr4}$ &  1.0 $\times$ 10$^{-4}$  &  (4)\\
o-D$_2$H$^+$ &  +  &gr$^-$  &$\rightarrow$&    2D   &+ &  H   \qquad\qquad +\quad   gr$^0$& 1/2 $k_{\rm gr5}$ & 1.3 $\times$ 10$^{-4}$   & (4)\\
\qquad&&\qquad              &$\rightarrow$&    o-D$_2$ & + &  H   \qquad\qquad +\quad    gr$^0$& 1/6 $k_{\rm gr5}$ & 4.4 $\times$ 10$^{-5}$   &  (4)\\
\qquad&&\qquad              &$\rightarrow$&    HD  & + &  D   \qquad\qquad +\quad   gr$^0$&1/3 $k_{\rm gr5}$ & 8.9 $\times$ 10$^{-5}$    &  (4)\\
p-D$_2$H$^+$ &  + & gr$^-$  &$\rightarrow$&    p-D$_2$ &  + &  H   \qquad\qquad +\quad    gr$^0$& 1/6 $k_{\rm gr5}$& 4.4 $\times$ 10$^{-5}$    & (4)\\
\qquad&&\qquad              &$\rightarrow$&    HD  & + &  D   \qquad\qquad +\quad   gr$^0$& 1/3 $k_{\rm gr5}$ & 8.9 $\times$ 10$^{-5}$    &  (4)\\
\qquad&&\qquad              &$\rightarrow$&    2D  & +&   H   \qquad\qquad +\quad    gr$^0$& 1/2 $k_{\rm gr5}$ &  1.3 $\times$ 10$^{-4}$  &  (4)\\
o-D$_3^+$ &  + &  gr$^-$    &$\rightarrow$&    3D  & + &  gr$^0$&   2/3 $k_{\rm gr6}$ & 1.6 $\times$ 10$^{-4}$   & (4)\\
\qquad&&\qquad              &$\rightarrow$&    o-D$_2$ &  +&   D   \qquad\qquad +\quad    gr$^0$& 1/3 $k_{\rm gr6}$ &  8.1 $\times$ 10$^{-5}$  &  (4)\\
m-D$_3^+$  & + &  gr$^-$    &$\rightarrow$&    3D   &+ &  gr$^0$&    2/3 $k_{\rm gr6}$ & 1.2 $\times$ 10$^{-4}$   & (4)\\
\qquad&&\qquad              &$\rightarrow$&    o-D$_2$ &  + &  D   \qquad\qquad +\quad    gr$^0$& 1/6 $k_{\rm gr6}$ &  4.1 $\times$ 10$^{-5}$  &  (4)\\
 \qquad&&\qquad             &$\rightarrow$&    p-D$_2$ &  + &  D   \qquad\qquad +\quad    gr$^0$& 1/6 $k_{\rm gr6}$ & 4.1 $\times$ 10$^{-5}$   &  (4)\\
p-D$_3^+$ &  + &  gr$^-$    &$\rightarrow$&    3D  & +& gr$^0$&    2/3 $k_{\rm gr6}$&  1.6 $\times$ 10$^{-4}$  & (4)\\
\qquad&&\qquad              &$\rightarrow$&   p-D$_2$  & +&   D  \qquad\qquad +\quad    gr$^0$& 1/3 $k_{\rm gr6}$ & 8.1 $\times$ 10$^{-5}$   &  (4)\\
HCO$^+$  &  + &  gr$^-$     &$\rightarrow$&    H   &+ &  CO   \qquad \quad\ +\quad    gr$^0$& $k_{\rm HCO^+}$ &  1.1 $\times$ 10$^{-4}$  & (4)\\
DCO$^+$  &  + & gr$^-$      &$\rightarrow$&   D & + & CO  \qquad \quad~+\quad \   gr$^0$& $k_{\rm DCO^+}$ & 1.1 $\times$ 10$^{-4}$ & (4)\\
N$_2$H$^+$ &   + &  gr$^-$  &$\rightarrow$&    H  & +  & N$_2$   \qquad\quad~ +\quad    gr$^0$& $k_{\rm N_2H^+}$ & 1.1 $\times$ 10$^{-4}$  & (4)\\
N$_2$D$^+$ &   + &  gr$^-$  &$\rightarrow$&    D &  +&   N$_2$  \qquad\quad~ +\quad   gr$^0$& $k_{\rm N_2D^+}$ & 1.1 $\times$ 10$^{-4}$ & (4)\\
\hline
\end{tabular}
\medskip
$^1$ $k'_{\rm CO}$ and $k'_{N_2}$ indicate that the $k_{\rm CO}$ and $k_{N_2}$ rates have been corrected for the higher reduced mass of the system because of the presence of \hddpb instead of \htp. Similarly, $k''$ and $k'''$ are given for \ddhpb  and \dtpb isotopologues.
\end{table}